Author's response to reviews

Title: Trends in the prevalence and management of diagnosed type 2 diabetes 1994-2001 in England and Wales

Authors:

Simon de Lusignan (slusigna@sghms.ac.uk)
Charalambos Sismanidis (Charalambos.Sismanidis@lshtm.ac.uk)
Iain M Carey (i.carey@sghms.ac.uk)
Stephen De Wilde (sdewilde@sghms.ac.uk)
Nicky Richards (NickyR@Compufile.co.uk)
Derek G Cook (d.cook@sghms.ac.uk)

Version: 3 Date: 14 March 2005

Author's response to reviews:

The formatting changes requested have been made.

There were no suggested reviewer changes to be made on the revision. (The only comment made was on 'missing' legend figures, but this was not relevant as your formatting requires that figure legends be kept separate from the figure)